Author’s response to reviews

Title: Comparison of device-based physical activity and sedentary behaviour following percutaneous coronary intervention in a cohort from Sweden and Australia: a harmonised, exploratory study.

Authors:

Nicole Freene (Nicole.Freene@canberra.edu.au)
Sabina Borg (sabina.borg@liu.se)
Margaret McManus (margaret.macmanus@act.gov.au)
Tarryn Mair (tarryn.mair@act.gov.au)
Ren Tan (ren.tan@act.gov.au)
Rachel Davey (raachel.davey@canberra.edu.au)
Birgitta Oberg (birgitta.oberg@liu.se)
Maria Back (maria.back@liu.se)

Version: 1 Date: 26 Jan 2020

Author’s response to reviews:

Please see the attached response to reviewers document.